Author's response to reviews

Title: Auditory change detection in schizophrenia: sources of activity, related neuropsychological function and symptoms in patients with a first episode in adolescence, and patients 14 years after an adolescent illness-onset.

Authors:

  Robert D. Oades (oades@uni-essen.de)
  Nele Wild-Wall (nele@wild-wall.de)
  Stephanie A. Juran (sjuran@web.de)
  Jan Sachsse (JanSachsse@web.de)
  Lyubov B Oknina (loknina@nsi.ru)
  Bernd Roepcke (bernd.roepcke@uni-essen.de)

Version: 5 Date: 7 February 2006

Author's response to reviews: see over
I have reordered some words in the title, removed the key words section, the headline-underlining and made minor changes to typos and punctuation in the text. I re-oriented the tables to portrait – as this required a change of font PLEASE CHECK that they appear normal before putting them on the web. With great difficulty I was able to re-adjust the references as required – I had to enlist the help of the BBF editor for this.